Author's response to reviews

Title: Prognostic significance of interim PET/CT based on visual, SUV-based, and MTV-based assessment in the treatment of peripheral T-cell lymphoma

Authors:

Sung-Hoon Jung (shglory@hanmail.net)
Jae-Sook Ahn (ahnjaesook@hanmail.net)
Yeo-Kyeoung Kim (kimcnuh@naver.com)
Sun-Seog Kweon (uijingogo@paran.com)
Jung-Joon Min (jjmin@jnu.ac.kr)
Hee-Seung Bom (hsbom@inu.ac.kr)
Hyeoung-Joon Kim (hjoonk@chonnam.ac.kr)
Yee Soo Chae (yschae@knu.ac.kr)
Joon Ho Moon (jhmoon@knu.ac.kr)
Sang Kyun Sohn (sksohn@knu.ac.kr)
Sang Woo Lee (swleenm@knu.ac.kr)
Byung Hyun Byun (dreamersball@paran.com)
Young Rok Do (dyr1160@dsmc.or.kr)
Je-Jung Lee (driejung@chonnam.ac.kr)
Deok-Hwan Yang (drydh1685@hotmail.com)

Version: 5 Date: 26 February 2015

Author's response to reviews: see over
Dear Editor-in-Chief

Please consider for publication our revised manuscript entitled “Prognostic significance of interim PET/CT based on visual, SUV-based, and MTV-based assessment in the treatment of peripheral T-cell lymphoma”. We have responded to the editorial request as described in the accompanied letter and revised a manuscript with changes highlighted. We hope that our revised manuscript will be acceptable for publication in the “BMC cancer”.

With best regards,

Deok Hwan Yang, M.D., Ph.D.
Department of Hematology-Oncology
Chonnam National University Hwasun Hospital
322 Seoyangro, Hwasun, Jeollanamdo 519-763, Republic of Korea
Tel: 82-61-379-7636
Fax: 82-61-379-7628
E-mail: drydh1685@hotmail.com
Editorial

1) "Please include in the Methods section the full name of the Institutional Review Board that approved the study"

(ANSWER). The full name of the Institutional Review Board that approved the study was described in the methods section as follows: All patients were eligible for inclusion after the protocol was approved by the Institutional Review Board of Chonnam National University Hwasun Hospital in accordance with the Declaration of Helsinki.